Author’s response to reviews

Title: Impact of early detection of acute invasive fungal rhinosinusitis in immunocompromised patients

Authors:

Mariana Silveira (dramacoelho@hotmail.com)
Wilma Anselmo-Lima (wtalima@fmrp.usp.br)
Francesca Maia (franmaia48@hotmail.com)
Danielle Queiroz (daniellecunha@ymail.com)
Rodrigo Nogueira (rodrigoln@unimedjaboticabal.coop.br)
Marcelo Leite (mjleiteorl@gmail.com)
Ricardo Lessa (dr.lessa918@gmail.com)
Belinda Simões (bpsimoes@fmrp.usp.br)
Edwin Tamashiro (edwin@fmrp.usp.br)
Fabiana Valera (facpvalera@fmrp.usp.br)

Version: 4 Date: 03 Oct 2018

Author’s response to reviews:

Re: BMC Infectious Diseases – INFD-D-18-00532R3 – Impact of early detection of acute fungal rhinosinusitis in immunocompromised patients

We are submitting a new version of the manuscript, in which we have completed deleted any direct comparison to our previous cohort.

We would like again to thank the Editor from his careful evaluation of our manuscript, and we hope that it can be now considered for publication in the BMC Infectious Diseases.

Thank you very much for this opportunity